Author's response to reviews

Title: A comparison of specialist rehabilitation and care assistant support with specialist rehabilitation alone and usual care for people with Parkinson’s living in the community: study protocol for a randomised controlled trial.

Authors:

Heather Gage (h.gage@surrey.ac.uk)
Sharlene Ting (s.ting@surrey.ac.uk)
Peter Williams (p.williams@surrey.ac.uk)
Karen Bryan (k.bryan@surrey.ac.uk)
Julie Kaye (juliemkaye@yahoo.co.uk)
Beverly Castleton (beverly.castleton@surreypct.nhs.uk)
Patrick Trend (ptrend@royalsurrey.nhs.uk)
Derrick Wade (derrick.wade@noc.nhs.uk)

Version: 5 Date: 16 November 2011

Author's response to reviews: see over
Dear Editors,

Thank you for writing to advise your acceptance of the above manuscript.

We have amended the manuscript in the light of the formatting requests.

- The title has been re-styled to conform to the journal style for study protocol articles:

  A comparison of specialist rehabilitation and care assistant support with specialist rehabilitation alone and usual care for people with Parkinson’s living in the community: study protocol for a randomised controlled trial.

- Ethics approval and funding documents have been removed.

- Formatting of all sections listed and to order them correctly:

  This has been addressed by altering headings and adding keywords where required.

- Figures and tables should not be included in the main body of the manuscript:

  Instructions state that “smaller tables …can be pasted into the end of the document text file” which is what we have done.

Yours sincerely

Karen Bryan
Corresponding Author